Author's response to reviews

Title: Characterization of human gastric carcinoma-related methylation of 9 miR CpG islands and repression of their expressions in vitro and in vivo

Authors:

Yantao Du (duyantao@bjmu.edu.cn)
Zhaojun Liu (liuzhaojun@bjmu.edu.cn)
Liankun Gu (gulkun@163.com)
Jing Zhou (jane72@163.com)
Bu-Dong Zhu (zhubd@tom.com)
Jiafu Ji (jiafuj@hotmail.com)
Dajun Deng (dengdajun@bjmu.edu.cn)

Version: 2 Date: 6 March 2012

Author's response to reviews:

Dear Editor:

We find that six supplementary tables are not uploaded with the submitted manuscript and have to revise the manuscript.

I am sorry.